Author’s response to reviews

Title: Adult growth hormone deficiency treatment with a combination of growth hormone and insulin-like growth factor-1 (Increlex) resulting in elevated sustainable IGF-1 and IGF binding protein 3 plasma levels: a case report

Authors:

Eric R Braverman (Pathmedical@aol.com)
Abdalla Bowirrat (bowirrat@netvision.net.il)
Uma J Damle (udamle04@yahoo.com)
Swetha Yeldandi (swethayeldandi@gmail.com)
Thomas J H Chen (tjhchen@yahoo.com.tw)
Margaret Madigan (MargBetts@hotmail.com)
Mallory Kerner (mallory.kerner@gmail.com)
Stanley X Huang (xhuang56@hotmail.com)
Stella Savarimuthu (stella.savarimuthu@gmail.com)
Kenneth Blum (drd2gene@aol.com)

Version: 4 Date: 4 June 2010

Author’s response to reviews:

Dear Sir:

Please find our revised manuscript by Braverman et al. “Adult growth hormone deficiency treatment with a combination of growth hormone and insulin-like growth factor-1 (Increlex) resulting in elevated sustainable IGF-1 and IGF binding protein 3 plasma levels: a case report” for your review.

We have made the following changes:

1) Authors’ contributions: Added “All authors read and approved the final manuscript.”
2) Title: Changed format to read “Adult growth hormone deficiency treatment with a combination of growth hormone and insulin-like growth factor-1 (Increlex) resulting in elevated sustainable IGF-1 and IGF binding protein 3 plasma levels: a case report”
3) Affiliations: Added countries to the affiliation details.
4) Removed all highlighting from previous version.
5) Competing interests section precedes the consent section.
6) Changed underlined text to bold.
7) Ensured figures were formatted properly.

Looking forward to a favorable review.
Professor Kenneth Blum, PhD.